Author's response to reviews

Title: Stereotactic body radiation therapy for post-pulmonary lobectomy isolated lung metastasis of thoracic tumor: survival and side effects

Authors:

Weijie Xiong (xiongweijie427@163.com)
Qingfeng Xu (93224872@qq.com)
Yong Xu (xy868996@163.com)
Changjin Sun (cjsun@163.com)
Na Li (lina_anhuimu@163.com)
Lin Zhou (drzhoulin@163.com)
Yongmei Liu (lymi75@163.com)
Xiaojuan Zhou (zhongliu201203@163.com)
Yongsheng Wang (wangys75@gmail.com)
Jin Wang (jinwang593@yahoo.com.cn)
Sen Bai (tonybaisen@yahoo.com.cn)
You Lu (radyoulu@hotmail.com)
Youling Gong (gongyouling@gmail.com)

Version: 4 Date: 17 September 2014

Author's response to reviews: see over
Dear reviewer and editor,

Thanks for your comments. According to your opinions, we already modified our manuscript and we would like to answer them in details here.

**Reviewer 1**

*Minor*

#1. L.98: I think "after pneumonectomy" should be "after lobectomy".  
*We already changed the word in the revised MS.*

#2. P.14: There are many typos.  
*Yes, these typos were our mistakes. And we already changed these words in the revised MS.*

**Editor**

Ethics: Experimental research that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate. For retrospective ("chart") studies, access to the data must have been granted by an appropriate ethics committee, and a statement to this effect should be included in the
Methods section of the manuscript.

*According to your comments, we already added the essential statement in the revised MS.*

And we also highlight all changes we made. If there is any problem, please let us know.

Sincerely yours,

Authors